Author's response to reviews

Title: Transcutaneous electrical nerve stimulation attenuates CFA-induced hyperalgesia and inhibits spinal ERK1/2-COX-2 pathway activation in rats

Authors:

Junfan Fang (fangjunfan0223@163.com)
Yi Liang (liangyiwww@126.com)
Junying Du (dujunying0706@163.com)
Jianqiao Fang (fangjianqiao7532@163.com)

Version: 4 Date: 27 May 2013

Author's response to reviews: see over
May 27, 2013

Dear Dr. Tom Rowles:

Thank you very much for your hard-work in processing our manuscript entitled “Transcutaneous electrical nerve stimulation attenuates CFA-induced hyperalgesia and inhibits spinal ERK1/2-COX-2 pathway activation in rats” (MS Number: 2167240758853412)

We are very pleased to be asked to submit a revision, and we found the comments and suggestions of the reviewers to be extremely helpful. We used a professional language editing service. Some grammar and spelling errors in the original manuscript had been corrected. We responded point by point to each reviewer comments as list below.

We believe the comments and professional language editing service have significantly improved the quality of manuscript and made it publishable.

Thanks again for your reconsideration of our manuscript for publication in your journal. We look forward to your favorable decision.

Yours sincerely

Jianqiao Fang.
RESPONSE TO REVIEWERS’ COMMENTS

MS Number: 2167240758853412  Title: Transcutaneous electrical nerve stimulation attenuates CFA-induced hyperalgesia and inhibits spinal ERK1/2-COX-2 pathway activation in rats

Thank reviewers for constructive suggestions and comments. Our responses are below:

Comments

Editorial comments:

We still feel that your manuscript would benefit from some improvements being made to the style and grammar of written English. We would therefore strongly recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service.

Response: It is really a grateful suggestion. We used the professional language editing service of Edanz (No. C1305-25059-FangJunfan). Some grammar, style and spelling errors in the original manuscript had been corrected. We believe the comments and professional language editing service have significantly improved the quality of manuscript and made it publishable. Thanks again.